Accelerator Status Report

Run Schedule: Sept. 27, 2010 to December 23, 2010
Nov. 3, 2010
Injector    Energy 61.76 MeV
· (11/1) Wien was changed from 29 to 5 degrees at Hall A’s request.

· The Gun ramping time was reduced by ~50%. Now it takes about 10 minutes to turn on and ramp up the Gun. 

· (10/26) Vertical Wien was calibrated.

· (10/21) Pockel Cell was off for about three shifts. We do not know why yet. The halls have access to it. The signal will be added to the Alarm Handler.
· (10/21) GUN air pressure fault. The faulty air hose and fitting were replaced.
Accelerator   Linac Energy 549.00 MeV
· Optics is good and stable.
· (11/3) Water flow interlock bypassed for FC#1. Chime added to drop mode to beam sync.

· (11/2) Path length drifted and it was corrected.
· (11/2) After running Krest eight cavity phases were adjusted 10 degrees. When manually

      checked them we found one phase adjustment was wrong. 
· RF 2L15 had HV problems a few times. Two of the cavities were bypassed.
· (10/29) ARC6A tripped on DC overload (0.72 hrs.).

· (10/28) MYR9R03 had communication fault and EES fixed it (1.13 hrs.). 

· (10/27) XSEP2T tripped off (0.85 hrs.).
· Hall B is at the highest pass. If things start to change (BLM trips, Hall B beam

       unstable), send tune to Hall B dumplette to check the machine.
· (10/26) Arc8 box supply tripped off and the machine was recovered quickly.

· (10/23) Saturday swing shift the West Arc lost power briefly so the box supplies and some iocs went down. Beam delivery was interrupted for one and a half hours.

· MOMOD system for higher pass is down due to a faulty lock-in amplifier. The CW path length tool can be used until the MOMOD system is back online.

· (10/22) RF separators still trip off one in a while.

· (10/22) PSS BCM fault dropped the injector to power permit.

· Hall A arc dipole tripped off on earth leakage fault several times last week.

· (10/21) No Hall A tune beam structure. It was fixed. AMS cross point switcher had problems to display tune beam. Pete resolved the issue.

· (10/21) most 4ch BPMs did not indicate. B. Bevins fond a bug in the BPM software and fixed it.

Hall A---Pass 4, DVCS
· Hall A beam was fine. 
· FFB feed forward button is on the FFB control screen now. 
· (10/27) Hall A did Moller runs. We did not back out of the Compton.
· The Compton counts have deteriorated. (10/25) The optics on-call and Jay did some tuning but it was not successful.
· (10/22) Hall A was changed from pass 5 to pass 4. Pass 5 separator was left on.

Hall B ---Pass 5, Primex II
· Hall B beam was fine.
· Hall B monitored some independent signals which show that their beam has been stable. 
      So most likely their high trigger rates ware not caused by the beam.

Hall C--- Pass 1, QWeak 
· Hall C did not take much beam due to issues with its cryo target. They are planning to take beam this weekend.
· Be careful with setting the FSD destination. Hall C FSD vacuum node is no longer maskable. This change should help prevent us from making mask mistakes. 
· (10/30) IPM3P02B wiring was corrected. (Two wires somehow got disconnected.)

· (10/30) Hall C Compton rates were good enough for now. 

· (10/30) Hall C Moller quads are on. It was off for a while due to water leak.

· (10/30) Owl shift noticed that Hall C dipole did not d a full cycle. The magnet should be fully cycled otherwise the field may not be reproduced. 

· (10/29) The hall installed a manual flag target. RadCon will give us training on it from RadCon point of view. 

· Hall C found that there is 500Hz noise in their bam line causing the beam to have double peaks vertically. The experts have been investigating. 
· Hall C has been working on the target pump, hopefully the pump will be back online this weekend so the hall can use its cryo target.
